







# AcneCare Online

## Acne Care Online:

feasibility and full-scale RCT of an online intervention to support acne self-management

### V2 29/ Nov/2023

**SPONSOR:** Building 37, University of Southampton, University Road, Highfield, Southampton SO17 1BJ Tel: 02380 595058

**COORDINATING CENTRE:** University of Southampton, Primary Care Research Centre.

ISRCTN reference: HRA/IRAS reference number: Sponsor reference number: Funder reference number:

To be added 330774 86407 NIHR202852

### Protocol authorised by:

Name:	Miriam Santer and Ingrid Muller	Role:	Co-Chief Investigators
Signatures:	M. Saute	Date:	23/11/23
	stull		23/11/23
Name:	Linda Hammond	Role:	On behalf of sponsor
Signature:	Ll Harman	Date:	21/12/2023

#### MAIN STUDY CONTACTS

Co-Chief Investigators Professor Miriam Santer and Dr Ingrid Muller Primary Care Research Centre University of Southampton Aldermoor Health Centre Aldermoor Close Southampton SO16 5ST

Email: <u>m.santer@sotonac.uk</u> <u>i.muller@soton.ac.uk</u>

#### STUDY COORDINATION CENTRE

For general study and clinical queries e.g., participant queries, study supplies, data collection, please contact in the first instance:

Academic Programme Manager: Rosie Essery Trial Manager: Julie Hooper Trial Coordinator: Rebekah Le Feuvre

Address:

Primary Care Research Centre University of Southampton Aldermoor Health Centre Aldermoor Close Southampton SO16 5ST

Tel: 02380 590742 Email: <u>acnecare@soton.ac.uk</u>

#### SPONSOR

University of Southampton is the research sponsor for this study. For further information regarding sponsorship conditions, please contact the Head of Research Ethics and Clinical Governance Linda Hammond (rgoinfo@soton.ac.uk):

Address:

Research Ethics and Governance Office Head of Research Ethics and Governance University of Southampton Room 4069/Building 37 Highfield Campus Southampton SO17 1BJ Tel: 023 8059 5058

#### TRIAL MANAGEMENT GROUP

Taeko Becque, Trial statistician, University of Southampton Sophie Dove, public contributor Zina Eminton, Southampton Clinical Trials Unit Rosie Essery, University of Southampton Nick Francis, University of Southampton Gareth Griffiths, Southampton Clinical Trials Unit Kate Henaghan-Sykes, public contributor Julie Hooper, University of Southampton Sinead Langan, London School of Hygiene and Tropical Medicine Alison Layton, Harrogate and District NHS Foundation Trust Rebekah Le Feuvre, University of Southampton Paul Little, University of Southampton Jacqueline Nuttall, Southampton Clinical Trials Unit Mahendra Patel, University of Bradford Matthew J Ridd, University of Bristol Tracey Sach, University of Southampton Irene Soulsby, public contributor Beth Stuart, Queen Mary University of London Kim S Thomas, University of Nottingham Andrew Thompson, Institution Cardiff & Vale University LHB Sarah Williams, Solent NHS Trust Lucy Yardley, University of Bristol

### FUNDER

This study is funded by NIHR Programme Grant for Applied Research NIHR202852 with additional support from the NIHR Clinical Research Network Service Support Costs.

This protocol presents independent research funded by the National Institute for Health and Care Research (NIHR) under its Programme Grants for Applied Research Programme (NIHR202852). The views expressed are those of the author(s) and not necessarily those of the NHS, the NIHR or the department of Health and Social Care.

#### **Protocol Information**

This protocol describes the Acne Care Online study and provides information about procedures for entering participants. The protocol should not be used as a guide for the treatment of other non- study participants; every care was taken in its drafting, but corrections or amendments may be necessary.

#### Compliance

This study will adhere to the principles of Good Clinical Practice (GCP). It will be conducted in compliance with the protocol, in accordance with current Data Protection Regulations and all other regulatory requirements, as appropriate.

### **TABLE OF CONTENTS**

LIS	ST OF ABBREVIATIONS	6
1	TRIAL SYNOPSIS	7
2	SCHEDULE OF OBSERVATIONS AND PROCEDURES	11
3	LAY SUMMARY	12
4	BACKGROUND AND RATIONALE	13
5	AIM AND OBJECTIVES	14
6	STUDY DESIGN	15
6.1	OUTCOME MEASURES	15
6.2	DEFINITION OF END OF STUDY	17
7	SELECTION AND ENROLMENT OF PARTICIPANTS	17
7.1	INVITATION TO PARTICIPATE	17
7.2	INCLUSION CRITERIA	19
7.3	EXCLUSION CRITERIA	19
7.4	CONSENT	20
7.5	CONFIDENTIALITY AND DATA PROTECTION	22
7.6	RANDOMISATION AND MASKING	22
8	SAFETY	22
8.1	SAFEGUARDING	22
9	FEASIBILITY TRIAL	23
10	INTERNAL PILOT AND FULL-SCALE TRIAL	23
11	STUDY PROCEDURES	23
11.1	L STUDY PROCEDURES AND FOLLOW UP	23
11.2	2 DEVIATIONS AND SERIOUS BREACHES	24
11.3	3 WITHDRAWAL	24
12	INTERVENTION AND GROUP DETAILS	24
13	STATISTICS AND DATA ANALYSES	25
13.1	1 METHOD OF RANDOMISATION	25
13.2	2 SAMPLE SIZE	25
13.3	3 STATISTICAL ANALYSIS PLAN (SAP)	25
14	NESTED HEALTH ECONOMIC EVALUATION	27

15	NESTED PROCESS EVALUATION STUDY	28
15.1	QUANTITATIVE PROCESS STUDY	29
15.2	QUALITATIVE PROCESS STUDY	30
15.3	PROCESS EVALUATION ANALYSIS	30
16	REGULATORY	31
16.1	CLINICAL TRIAL AUTHORISATION	31
17	ETHICAL CONSIDERATIONS	31
17.1	ETHICAL APPROVAL	31
18	SPONSOR	31
18.1	INDEMNITY	31
18.2	FUNDING	31
18.3	AUDITS AND INSPECTIONS	32
19	STUDY OVERSIGHT GROUPS	32
19.1	PROGRAMME MANAGEMENT GROUP (PMG)	32
19.2	PROGRAMME STEERING COMMITTEE (PSC)	32
19.3	DATA MONITORING AND ETHICS	32
20	RECORD RETENTION AND ARCHIVING	32
21	PUBLICATION POLICY	33
22	REFERENCES	33

### LIST OF ABBREVIATIONS

Acne-QoL	Acne-Specific Quality of Life questionnaire
Acne vulgaris	Acne
CEAC	Cost Effectiveness Acceptability Curves
CONSORT	Consolidated Standards of Reporting Trials
DMEC	Data Monitoring and Ethics Committee
GCP	Good Clinical Practice
GP	General Practice or General Practitioner
HEAP	Health economics analysis plan
ICER	Incremental Cost-Effectiveness Ratio
IPQ	Illness Perceptions Questionnaire
MCID	Minimum Clinically Important Difference
MHRA	Medicines and Healthcare products Regulatory Agency
NHS	National Health Service
NIHR	NHS Institute for Health & Care Research
PEI	Patient Enablement Instrument
PETS	Problematic Experiences of Therapy Scale
PHQ-4	Patient Health Questionnaire
REC	Research Ethics Committee
PMG	Programme Management Group
PSC	Programme Steering Committee
QALY	Quality Adjusted Life Years
QoL	Quality of life
RCT	Randomised Controlled Trial
SAP	Statistical analysis plan
SWEMWBS	Short Warwick Edinburgh Mental Well-being Scale
UoS	University of Southampton

### **Keywords**

Acne, primary care, community care, pharmacy, schools, digital intervention, adherence, selfmanagement, randomised controlled trial, process evaluation, economic evaluation

### 1 TRIAL SYNOPSIS

<b></b>			
Short title/Acronym:	Acne Care Online		
Full title:	Acne Care Online: feasibility and full-scale RCT of an online intervention to support acne self-management		
Population:	Young people with acne (aged 13-25 years)		
Objective:	To determine the feasibility and effectiveness of an online behavioural intervention aimed at supporting self-management, improving outcomes and reducing antibiotic use in acne		
Study Designs:	A multi-centre randomised controlled trial with a feasibility trial and full-scale randomised trial with internal pilot, nested health economic evaluations and mixed-methods process evaluation studies		
Sample sizes:	Feasibility study: 60 participants		
	Feasibility qualitative process interviews: up to 10 participants from each trial arm (20 maximum)		
	Full-scale trial: 588 participants (including 85 in internal pilot)		
	Full scale trial qualitative process interviews: 20-25 participants		
Intervention:	Access to online behavioural intervention for people with acne to support effective acne self-management, in addition to usual care		
Control group:	Usual care, including signposting to NHS advice from baseline and access to the intervention after 52 weeks follow-up		
Follow up duration:	52 weeks		
Recruitment sites	Feasibility trial, internal pilot and full-scale trial will recruit through general practices, community pharmacies, social media and other community recruitment, such as schools and colleges.		
	<b>Feasibility study:</b> We anticipate needing 8-12 GP sites for the feasibility trial to randomise 30 participants (half of total as the remainder to be recruited through social media and community advertising). The exact number of sites will be determined by recruitment rates.		
	<b>Full-scale study with internal pilot:</b> We anticipate needing around 100 GP sites for the full-scale trial to randomise 300 participants (half of total as the remainder to be recruited through social media and other sources). The exact number of sites will be determined by recruitment rates.		

URL for randomisation:	www.lifeguideonline.org			
Outcome measures	Both feasibility trial and full-scale trial will include the following measures:			
	Primary outcome			
	Acne severity measured using the Acne-QoL symptoms subscale at 12 weeks, compared with baseline			
	Other measures (including secondary outcomes, process measures and economic measures)			
	Acne severity evaluated over 12 months using Acne-QoL symptom subscale, using repeated measures analysis over 12 weeks, 24 weeks, 36 weeks and 52 weeks, controlled for baseline			
	Acne-QoL other subscales (self-perception, role-emotional and role-social) and total score evaluated over 12 months using Acne-QoL symptom subscale using repeated measures analysis over 12 weeks, 24 weeks, 36 weeks and 52 weeks, controlled for baseline			
	Use of topical treatment for acne at baseline, 12 weeks, 24 weeks, 36 weeks and 52 weeks (self-report and treatment adherence)			
	Use of acne-related antibiotics and other oral acne treatment use at baseline, 12 weeks, 24 weeks, 36 weeks and 52 weeks, including frequency of use (self-report)			
	Past use of topical and oral treatments for acne			
	Problematic Experiences of Therapy Scale (PETS) will be asked prior to adherence questions at baseline,12 weeks and 52 weeks.			
	Patient Enablement Instrument (PEI) at baseline, 12 weeks and 52 weeks			
	Brief Illness Perceptions Questionnaire (IPQ) at baseline, 12 weeks and 52 weeks			
	Patient Health Questionnaire (PHQ–4) for Anxiety and Depression at baseline, 12 weeks and 52 weeks			
	EQ-5D-5L at baseline, 12 weeks, 24 weeks, 36 weeks and 52 weeks			
	Short Warwick Edinburgh Mental Well-being Scale (SWEMWBS) at baseline, 12 weeks, 24 weeks, 36 weeks and 52 weeks			
	Resource use will be measured at baseline, 12 weeks, 24 weeks, 36 weeks and 52 weeks (self-report)			

	<ul> <li>Prior belief in effectiveness of online interventions will be asked at baseline and use and type of online resources previously used for acne will be asked at baseline, 12 weeks and 52 weeks</li> <li>Demographics</li> <li>Age of onset and duration of acne</li> <li>We will assess questionnaire burden in the feasibility trial and if this is not acceptable to participants we will consider not including the Brief IPQ and SWEMWBS in the full-scale trial and internal pilot</li> </ul>		
Statistical methods	Feasibility trial:		
	Descriptive statistics will describe recruitment and retention rates, completion of questionnaire items and characteristics of participants. All outcome measures will be included in the feasibility trial to assess acceptability of trial processes and questionnaire burden.		
	Full-scale trial sample size:		
	Based on comparison of the Acne-QoL symptoms subscale at 12 weeks, power 90%, alpha 0.05 and seeking a standardised effect size of 0.30 (standard deviation 5.8), 470 participants are needed. Allowing for 20% loss to follow up gives a total 588 participants (294 per arm).		
	Full-scale trial analysis:		
	The primary analysis will be a linear regression model to analyse Acne-QoL symptom subscale at 12 weeks, adjusting for key baseline covariates and stratification factors.		
	As a secondary analysis we will also assess Acne-QoL symptom subscale and other Acne-QoL subscales with a repeated measures approach, using a multilevel mixed model and adjusting for the same covariates as the primary analysis.		
	Secondary outcomes will be modelled using a regression modelling approach with a distribution appropriate to the outcome. All analyses will be on an intention to treat basis.		
Process evaluation	Feasibility trial:		
	We will conduct up to 10 qualitative interviews with participants in the intervention group and up to 10 participants in the control group to explore facilitators and barriers to trial recruitment and retention and to inform programme theory for design of the process evaluation in the full-scale trial. <b>Full-scale trial:</b>		

	Qualitative interviews will be carried out with 20-25 trial participants to provide in-depth understanding of experiences and factors that influence engagement with the intervention. This will include approximately 15-20 people in the intervention group and approximately 5 people in the control group. Quantitative process evaluation will assess baseline data (age, gender, acne severity) to examine moderator effects on intervention engagement (automated intervention usage data and self-reported treatment adherence) and outcomes. We will assess potential mediators of intervention adherence with outcomes; changes in questionnaire items on beliefs about treatment (PETS) and enablement (PEI), as well as intervention usage.		
Economic evaluation	<ul> <li>Feasibility trial:</li> <li>All economic measures will be included in the feasibility trial to assess completion rates and acceptability of trial processes.</li> <li>Full-scale trial:</li> <li>A within-trial cost utility analysis comparing the online intervention to usual care will be undertaken from an NHS perspective. Resource use will be collected via self-report questionnaires at baseline, 12 weeks, 24 weeks, 36 weeks and 52 weeks. Health-related quality of life will be measured using EQ-5D-5L scores at the same timepoints, which will be converted to utility scores to estimate Quality-Adjusted Life years for the trial period. Cost utility analyses using regression analysis will</li> </ul>		
	estimate the incremental cost, incremental benefit and incremental cost effectiveness of the online intervention compared to usual care alone. Sensitivity analyses will be undertaken to explore key uncertainties. If the inclusion of the Short Warwick Edinburgh Mental Well- being Scale (SWEMWBS) proves acceptable in the feasibility study we will estimate Mental Well-being Adjusted Life Years (MWALY).		

Outcomes collected	Baseline	12 weeks	24 weeks	36 weeks	52 Weeks (end of study)
Sign up information required	х				
Screening questions	х				
Demographics	х				
Age of onset and duration of acne	х				
Acne-QoL	х	х	х	x	x
Current use of topical acne treatments (which one)	х	x	х	x	x
Past use of topical acne treatments (which one)	х				
Problematic Experiences of Therapy Scale (PETS)	х	x			x
Adherence to topical acne treatments (frequency of application)	х	x	x	x	х
Current use of oral acne treatments (which one)	х	x	х	x	x
Past use of oral acne treatments (which one)	х				
Patient Enablement Instrument (PEI)	х	x			x
Prior belief about effectiveness of intervention	х				
Use of online resources	х	х			x
Patient Health Questionnaire (PHQ–4)	х	x			x
EQ-5D-5L	х	х	х	х	х
Short Warwick Edinburgh Mental Well-being Scale (SWEMWBS)	х	x	x	x	x
Brief Illness Perceptions Questionnaire	х	x			x
Resource use	х	х	х	х	x

### 2 SCHEDULE OF OBSERVATIONS AND PROCEDURES

Data collection during intervention period is kept to a minimum to reduce participant burden and reduce chance of regular monitoring over-shadowing intervention effects. Total questionnaire duration is 10 minutes and has been assessed as acceptable by public contributors in the target age range.

### **3 LAY SUMMARY**

### Background

Acne is very common, frequently causes distress or low self-confidence, and may lead to permanent scarring, long-lasting dark marks, or depression. Treatment of acne is a major cause of antibiotic use amongst young people, leading to antibiotic resistance. Evidence and guidelines suggest that topical treatments (creams or gels applied directly to the skin) should be the main treatments for acne. Effective topical treatments are available from pharmacies without a prescription, but many people are unaware of these and buy cosmetic products that don't help instead. People often give up on topical treatments because they are not given full advice on how to use them. For example, they don't know how to reduce the risk of stinging and redness or that it takes several weeks for treatments to start working.

We have developed a new website to help young people with acne to manage acne more effectively, including information on how to obtain effective treatments, promote regular treatment use and advice on how to avoid side effects.

### Aims

We will test the new website in a randomised trial to see whether it improves outcomes for people with acne and reduces the use of long-term oral antibiotics.

#### Methods

People with acne (aged 13-25 years) will be invited to take part in the study through a variety of sources including from GP practices, pharmacies, community and social media advertising, and through schools and other partner organisations. Participants will be asked to register online and complete an online consent form and questionnaire before being allocated by chance (randomly) to either the new study website, or to the control group. All participants will be able to access their usual health care or acne treatments during the study. Participants will be asked to complete questionnaires at 12 weeks, 24 weeks, 36 weeks and 52 weeks asking about their acne, quality of life and their use of acne treatments.

Participants in the control group will be signposted to standard NHS advice about acne. They will also be offered access to the new website after 52 weeks follow-up.

We will also explore how the new website works, by talking to those that have used it and analysing how they used it, and explore the costs to the NHS and for patients of using the website and estimate its value for money.

### Dissemination

We will publish this work in peer-reviewed journals, at conferences and via patient organisations. If effective, we will aim to make the new website freely available and ensure widespread uptake by working with young people and health care providers to signpost and endorse its use.

### 4 BACKGROUND AND RATIONALE

Acne vulgaris (hereon acne) is extremely common, affecting over 90% of teenagers.[1] It is primarily a condition affecting the pilosebaceous unit (hair follicles in the skin and associated oil glands), leading to inflammatory and non-inflammatory lesions and scarring,[2] most commonly affecting the face. The aetiology remains unclear but research into dietary, hygiene or other lifestyle causes have shown these to have little influence.[2]

Acne can cause significant distress, decreased self-confidence and increased rates of depression and suicidal ideation.[2] Distress about the appearance of acne appears to be particularly high amongst women and people identifying as non-white.[3] The majority of acne is classed as mild, with moderate or severe acne combined making up less than 15% of acne.[4, 5] Acne persists in 40-60% of people into their twenties and is thought to lead to some degree of scarring in approximately 20% of the population.[2, 4]

As well as limiting quality of life, acne makes a major contribution to antibiotic use amongst young people,[6] results in substantial health service use[7] as well as causing skin changes that can be permanent, such as skin pigment changes and scarring.[2] Skin conditions account for 18% of antibiotics prescribed outside hospitals[8] and, although data are scarce, antibiotic resistance arising from acne treatment is a major and increasing concern.[9] Health service use for acne is substantial with approximately 3% of people aged 13-25 visiting their GP for acne each year[7] (10k consultations per year in the UK), of whom 8% are referred to secondary care.[7] The majority of people attending secondary care are prescribed isotretinoin and require frequent review (monthly for women).[10]

There is high-quality evidence supporting use of topical treatments for acne, which are recommended as first line treatment for mild-to-moderate acne in UK guidelines.[11] However, treatment adherence is low, mainly due to limited understanding, with insufficient motivation and support necessary to engage effectively with treatments (regular use is needed over several weeks) and the barrier of skin irritation unless mitigating advice is emphasised.[12] These problems can be overcome with better support.[13]

Another major barrier to the use of effective treatment is a perception that acne is not a problem that warrants medical attention, even where the condition has considerable emotional and social impact.[14, 15] People with acne often have little awareness of effective topical treatments available via pharmacy or on prescription and instead use ineffective cosmetic products.[16] They frequently consult late, and at a point when they are disillusioned with topical treatments and are keen for oral treatments, commonly antibiotics, which are incorrectly perceived as being more effective than topical treatments.[16] GPs prescribe oral antibiotics during 31% of first consultations for acne,[17] with median duration 4 months,[6] partly due to perception of high patient demand for antibiotics.[18]

NHS England describes acne as a condition appropriate for self-management with pharmacy advice and use of over-the-counter medicine.[19] Accessible support is needed for people with acne to gain early control of the condition through helping them obtain effective treatment, use this regularly and avoid side effects. Early treatment of acne is a priority in order to avoid oral antibiotic use, lessen psychosocial impact and reduce risk of severe acne or scarring that

may lead to further treatment, referral and psychological consequences,[9] as early treatment can minimise subsequent scarring and associated referrals.[20]

Although there have been many calls for improved education and information for people with acne, [2, 12, 13] there is little research evaluating behavioural interventions to date, and education and information are insufficient to change behaviour effectively: the most recent review identified four RCTs of interventions; two used emails or text messaging (both small and inconclusive) and two used increased follow-up visits.[21]

A recent study of an instructional video on how to use topical treatments for acne, delivered via a pharmaceutical company app, found weak evidence of increased treatment adherence, but there were methodological limitations and the app is not widely used.[22] An industry-funded guideline has been developed for pharmacists but not widely used.[23]

Digital interventions change health-related behaviours when based on robust development supported by theory and multiple behaviour techniques.[24] A Cochrane review has shown that digital interventions improve outcomes in long-term conditions.[25] Given the evidence above for behavioural barriers to treatment (i.e. low awareness of effective topical treatments, low adherence and insufficient management of side effects), there is a good case for the development of a robust intervention to support effective self-management of acne. We focus on young people aged 13-25 years with mild and moderate acne, as this is the largest group, most likely to benefit and because acne in people over the age of 25 and severe acne can be more difficult to manage.

An intervention to improve self-management for acne is likely to have the following potential benefits:

- Effective early management of acne may reduce the emotional, psychological, and social burden of acne;
- Effective early management of acne may reduce the risk of sequelae such as skin pigment changes or scarring, for which there are few effective treatments;
- Improved use of topical treatments is likely to reduce oral antibiotic use, reducing the risks from antimicrobial resistance.

### 5 AIM AND OBJECTIVES

### Aim

To improve outcomes for acne through an online behavioural intervention to help people selfmanage acne effectively, including by promoting appropriate treatment use and reducing overuse of long-term oral antibiotics

### Objectives

- Assess the feasibility of conducting a randomised trial of an online self-management intervention for acne.
- Evaluate clinical effectiveness of online self-management intervention for acne in improving acne outcomes

- Explore mechanisms underlying effectiveness of the intervention through process evaluation
- Undertake an economic evaluation to estimate cost-effectiveness of the intervention compared with usual care from NHS and patient perspectives

### 6 STUDY DESIGN

This study is part of a programme of work which has developed an online intervention to support self-management for acne to be evaluated in a feasibility and full-scale, 52 week randomised controlled trial (RCT) with nested health economic and process evaluation studies.

The intervention includes online information and support for young people with help-seeking from healthcare professionals for acne treatments and advice, including a simple patient-held 'decision tool' that people with acne can take to consultations with GP or community pharmacy professionals, in order to support communication within the consultation. Participants will be recruited via GPs, community pharmacies, social media and other community recruitment, such as schools and colleges. We will recruit via all these routes for both feasibility trial and full-scale trial, as assessing rates of recruitment and diversity of recruitment is a key feasibility outcome.

Participants in both feasibility trial and full-scale trial will be randomised to one of two groups:

- 1. Usual care plus immediate access to the online intervention
- 2. Usual care with signposting to NHS advice and access to the online intervention after 52 weeks of follow-up

### 6.1 OUTCOME MEASURES

All participant-reported outcome measures and intervention usage data will be collected online for both the feasibility and full-scale trials. This will be collected using LifeGuide software (<u>www.lifeguideonline.org</u>), which is a unique set of open source software used to develop and trial online behavioural interventions.

Feasibility trial will include the following measures as it is anticipated that these will all be required for the full-scale trial in order to allow for process evaluation and economic evaluation, and acceptability of completion needs to be assessed at feasibility stage. Preliminary public contributor input has suggested that questionnaire length is acceptable (approximately 10 minutes). However, we will assess questionnaire burden in the feasibility trial and if this is not acceptable to participants we will consider reducing these for the fullscale trial and internal pilot.

### 6.1.1 Primary outcome measure

The primary clinical outcome measure is acne severity measured using the Acne-QoL symptoms subscale at 12 weeks. Although our previous research shows there are several acceptable participant-reported outcome measures for acne, [26] Acne-QoL is the most extensively validated. [27-29] There is not yet a core outcome set developed for acne, although the Acne-QoL is compatible with current developments. [30, 31]

The effect of acne treatments is seen around 6-12 weeks,[32] hence our choice of primary timepoint at 12 weeks. Longer term follow-up to 52 weeks will also be important in order assess the impact of the intervention on maintenance as well as initial control of acne, as relapse in acne following discontinuing treatment is thought to contribute to consulting and lack of acne control.

### 6.1.2 Other data and outcomes

The further measures to be completed by online questionnaire are as follows. Some are both secondary outcomes and process measures (such as adherence to topical treatments) while some are both process measures and measures to inform cost utility (such as Short Warwick Edinburgh Mental Well-being Scale) so we therefore list them as 'other measures' rather than separating by secondary outcome, process measure or economic outcome.

- Acne severity will be assessed over 12 months by measuring Acne-QoL symptom subscale, using repeated measures analysis over 12 weeks, 24 weeks, 36 weeks and 52 weeks, controlled for baseline
- Acne-QoL other subscales (self-perception, role-emotional and role-social) and total score, using repeated measures analysis over 12 weeks, 24 weeks, 36 weeks and 52 weeks, controlled for baseline
- Demographics will be asked at baseline: age, gender, ethnicity, postcode, occupation of main household earner
- Prior belief in effectiveness of online interventions for acne will be asked at baseline
- Use and type of online resources previously used for acne will be asked at baseline, 12 weeks and 52 weeks
- Age at onset of acne and duration of acne will be asked at baseline
- Current use of topical and oral treatments for acne will be asked at baseline, 12 weeks, 24 weeks, 36 weeks and 52 weeks
- Past use of topical and oral treatments for acne will be asked at baseline
- Adherence to topical acne treatments (frequency of application) will be asked at baseline, 12 weeks, 24 weeks, 36 weeks and 52 weeks
- Problematic Experiences of Therapy Scale (PETS)[33] will be asked prior to adherence questions at baseline, 12 weeks and 52 weeks. The PETS asks participants to what extent they had been prevented from carrying out the intervention by socially acceptable reasons in order to explore barriers to adherence and promote reporting of non-adherence (e.g. "Lack of time prevented me from carrying out the treatment")
- Patient Enablement Instrument (PEI)[34] will be measured at baseline, 12 weeks and 52 weeks to assess the extent to which usual care or intervention use has enabled participants to understand and self-manage their acne
- Brief Illness Perceptions Questionnaire (IPQ)[35] will be measured at baseline, 12 weeks and 52 weeks to explore participants beliefs about their acne that are expected to be important determinants of adherence to treatments (e.g. perceived impact, severity, control, treatability, timeline)
- Patient Health Questionnaire (PHQ-4) for Anxiety and Depression[36] will be measured at baseline, 12 weeks and 52 weeks

- EQ-5D-5L [37] will be used as a measure of generic quality of life (QoL) at baseline, 12 weeks, 24 weeks, 36 weeks and 52 weeks
- The Short Warwick Edinburgh Mental Well-being Scale (SWEMWBS) will be elicited at baseline, 12 weeks, 24 weeks, 36 weeks and 52 weeks, and valued using a UK preference-based value set to estimate Mental Well-being Adjusted Life Years (MWALY). [38] In a previous acne trial (SAFA),[39] a large proportion of the participants started the trial in perfect health according to the EQ-5D (46% in the intervention group and 43% in the control group) and thus the EQ-5D had no capacity to measure any benefit that might occur from the intervention. There has also been some debate about how well the EQ-5D captures mental well-being given 4 of the 5 domains measure physical health. Since acne can significantly affect mental-wellbeing we aim to test and compare the performance of SWEMWBS to EQ-5D as a generic measure of quality of life for people with acne.
- Healthcare resource use will be measured at baseline, 12 weeks, 24 weeks, 36 weeks and 52 weeks – data collection will be kept minimal, and the online format will allow for skipping of sections where no resources have been used. We intend to collect this information every 3 months, with a simple diary prompt, in order to avoid asking participants to recall over longer periods. Self-reported health care resource use has been found to provide acceptable data across other studies.[40, 41]

We will ask about the primary outcome, treatment use, resource use and generic QoL at all time points. We will ask about intervention effects, i.e., treatment adherence and barriers to adherence (PETS), enablement (PEI) and other secondary outcomes (PHQ-4) just at primary timepoint (12 weeks) and 52 weeks, to assess whether there has been maintenance of intervention effect.

### 6.2 DEFINITION OF END OF STUDY

End of full-scale RCT is defined as the date when the last point of data is collected for the last participant from their follow-up questionnaire.

### 7 SELECTION AND ENROLMENT OF PARTICIPANTS

### 7.1 INVITATION TO PARTICIPATE

Recruitment and enrolment procedures, and inclusion criteria will be the same for both feasibility and full-scale trials, so that feasibility of recruitment can be assessed prior to full-scale trial, and necessary adaptations made.

Young people aged 13-25 with acne will be recruited through a variety of sources including mail-out and opportunistic recruitment from general practices, opportunistic recruitment in pharmacies, community and social media advertising, and through schools, colleges, pharmacies and other partner organisations. We will develop age-appropriate recruitment materials to explain what is involved for potential participants.

### 7.1.1 General practice recruitment

Invitations will be sent through SMS text and/or postal mail out, poster advertising (16-25s only) and/or opportunistic recruitment by participating GP practices.

#### SMS/postal mailout:

Database searches will identify patients aged 13 to 25 years who either:

- Have been prescribed a topical treatment for acne or co-cyprindiol in the previous 12 months or
- have a diagnostic code for acne on their electronic record over the previous 12 months either with or without having been prescribed an oral tetracycline or erythromycin in that period.

GPs will be asked to screen the lists to check for patients who it may be inappropriate to invite, such as known opposition to taking part in research. Patients aged 16 and over will be contacted directly whereas for those under the age of 16, initial contact will be made with their parent/guardian.

Potential participants (or their parent/guardian where under 16) will be sent the following: 1) SMS text message to their registered mobile number with a link to the study website; and/or 2) postal mail out pack including a summary participant information sheet (PIS) and details of how to access the study website. The study website will include full participant information and parental information, in both accessible (summary PIS) and detailed formats (full PIS), who to contact for further information or discussion, information on eligibility and how to proceed to provide online consent prior to participating in study activities.

Discussions with public contributors have suggested that their parent/carer may not remember to forward messages on so it would be preferable for the invitation to be sent by more than one method, such as by text and by post. We therefore propose to use both routes, inviting participants by text or post or both if necessary. We will not send more than one text invite or more than one postal invite to each potential participant (or their parent/guardian where under 16).

Discussions with GP practices indicate that is not always possible to ascertain whose mobile phone number or email address is linked to GP records. Both public contributors and GP practices have said that, for young people aged over 16, it is often the parent/carer's mobile number still listed as the patient's 'preferred' mobile number, as they have not yet changed this. In these instances, the initial invitation may go to the parent rather than the young person in error, although subsequent links will request the parent/carer to forward the text to the young person so that they can access the participant information sheet and find out how to proceed to provide online consent prior to participating in study activities.

Opportunistic recruitment in general practice will also be offered to potential participants (or their parent/guardian where under 16). The health professional seeing the patient will then send an SMS text message to their mobile number with a link to the study website.

### 7.1.2 Opportunistic recruitment in pharmacies

Potential participants, or their parent/carer if under 16, will be given a link (via advertisement on screens (16-25 only) or by summary participant information sheet given by the pharmacy professional) directing them to the study website. The study website will include full participant information sheet, who to contact for further information or discussion, information on eligibility and how to proceed to provide online consent prior to participating in study activities.

### 7.1.3 Recruitment through schools/colleges

We will engage with teaching, pastoral or school nursing staff to plan opportunities to raise awareness of the research, including the possibility of research participation and how to take part. This may include invitations to take part in the study via letter, SMS text or email from the school or college directing interested individuals to the study website. All potential participants invited via this route will be contacted initially via their parent/carer given that many are likely to be under the age of 16.

### 7.1.4 Community or social media advertising

Community or social media advertising will be targeted at potential participants aged 16 or over. Posters will be displayed in locations such as GP surgeries, hospitals, community pharmacies, post-16 education colleges, University campuses, local newspapers or social media inviting potential participants, directing them towards the study website, which will include full participant information, who to contact for further information or discussion, information on eligibility and how to proceed to provide online consent prior to participating in study activities.

### 7.2 INCLUSION CRITERIA

Participants will be eligible for inclusion if:

- They are aged 13-25 years with self-defined acne with current active lesions (i.e. mild or worse on self-assessment scale)
- Have internet access

### 7.3 EXCLUSION CRITERIA

Potential participants will be excluded if:

- Their acne is currently clear or almost clear, i.e. if they answer 'no' to the following question asked online at baseline screening: "Do you have spots or acne on your face at the moment? This could include whiteheads or more red spots or bumps."
- They are unable to give informed consent or their parent does not provide consent (for participants aged 13-15 recruited via community or social media advertising)
- They are unable to read and write English as the intervention content and outcome measures are in English
- They are currently taking oral isotretinoin or have taken it within the previous 3 months, as advice about topical acne treatments may be inappropriate in this case

- They took part in interviews as part of Acne Care Online intervention development. (Qualitative interviewees who did not view intervention materials will NOT be excluded)
- Participants in the feasibility trial will be excluded from the full-scale trial
- Only one person per household will be able to take part in the study. If someone from that household has already joined the study then they will be excluded from the study

### 7.4 CONSENT

Participants will complete informed consent online through study software, prior to completion of baseline questionnaires and randomisation. The right of the participant to refuse to participate without giving reasons will be respected. All participants are free to withdraw at any time from the trial without giving reasons and without prejudicing further treatment. The consent form will include a question requesting explicit consent to access relevant data on intervention usage.

Our procedures are aimed at striking a balance between enabling young people to take part in research, as young people are currently underserved in acne research, vs ensuring informed age-appropriate consent to participate in research.

### Participants aged 16 or over

Consent to enter the trial will be sought online from each participant only after they have received full study details and study team contact information (telephone number and email address), to allow time for consideration and the opportunity to ask questions.

### Participants aged 13 to 15 recruited via their registered general practice or school

The initial invitation for young people aged 13 to 15 will be sent to their parent/carer, either through their registered practice or their school. They will be provided with a parental information sheet, or link to the study website where they can review this information. This will advise them what the study would involve for their child and will provide the opportunity to contact the research team to ask any questions. If they choose to pass this invitation on to the young person this will mean that 'implied consent' has been given. This would mean that a young person under the age of 16 would not receive materials inviting them to participate in the study unless their parents choose to pass on this information. The young person will then have access to the study website where they can review full study details and study team contact information (telephone number and email address), read the PIS, and will have the opportunity to request further information/ have questions answered. Online informed consent will then be obtained from participants aged 13-15.

In previous studies, where the parent/carer has provided consent and provided contact details for the young person there has been a very high drop-off in number of people engaging with the study. It is unknown whether this is because the young people are not interested in the study, or whether they are uninterested in research presented to them by their parents, but the result is under-representation of under-16s in research. Including this process of 'implied consent' from parents/carers promotes accessibility to research participation for young

people. This is particularly appropriate for a low risk study where all trial procedures are online and the decision to be randomised only relates to choice between new online information and standard NHS information.

Participants aged 13 to 15 recruited via community advertising, social media or pharmacy Community or social media advertising will be targeted at potential participants aged 16 or over only. However, if a young person aged 13-15 wishes to participate (for instance having been given the link by a friend or sibling), then they will receive an explanation that their parent/carer needs to provide consent for them to participate. They will be asked to direct their parent/carer to the study website, which includes full parent information sheet, who to contact for further information or discussion, information on eligibility and how to take part. If a parent provides consent for their young person aged 13-15 to take part then the young person will also be asked to provide consent.

This differs from the process of 'implied consent' for schools and GPs, where it can be ensured that the initial invitation goes to the parent/carer, who then passes it on to the young person, thus forming implied consent. This process cannot be ensured for social media and community advertising, hence the need for parent/carer consent.

#### **Process evaluation interviews**

For both feasibility and full-scale trials, participants will be invited to take part in a qualitative study (as part of the process evaluation), and will be asked at baseline whether they consent to be contacted about the interviews. When the interviews are due, participants will be contacted by phone, text message or email to give them further information about the interview study and the opportunity to ask questions. They will then be asked to login to give their consent online or provide written consent prior to the interview, or if they do not manage to complete either of these but attend for an interview then verbal consent will be recorded at the start of the interview. Where face-to-face interviews are carried out written informed consent will be sought in person. Participants will be free to withdraw at any time by contacting the research team.

Interviews will be carried out by experienced qualitative interviewers, either face-to-face or remotely via videoconferencing (Teams) or telephone. Face-to-face interviews will be held in participants' homes, but if they would prefer to meet elsewhere, alternative arrangements may be made for the meeting to be held on University Premises. Encrypted audio-recorders will be used for any interviews that do not take place on Microsoft Teams. Audio recordings will be professionally transcribed, anonymised and checked.

For process evaluation interviews with participants aged 13-15, consent from their parent/carer will be obtained prior to seeking consent from the young person. Feedback from public contributors suggests that 'implied consent' for the trial is acceptable for a low risk trial carried out entirely online, but that parental consent should be obtained in order for the researcher to speak to a research participant aged under 16.

### 7.5 CONFIDENTIALITY AND DATA PROTECTION

The confidentiality of all participants taking part in the study will be preserved. The study team will ensure that participants' anonymity is maintained and that their identities are protected from unauthorised parties. They will not be identified by their names, but by an identification code. Transcribers of the qualitative process evaluation interviews will sign a confidentiality agreement. The encrypted audio-recorders will be stored in locked offices at the University of Southampton.

Participant data will be collected online via LifeGuide software and stored securely on servers hosted by the University of Southampton. The participant data is pseudo anonymised by assigning each participant a participant identification number which will adapted for use in LifeGuide. Any follow-up data collected by telephone will be entered remotely at site. For qualitative interviews, all electronic data will be stored on a secure server until the transcriptions have been completed. Once these have been carried out and unique identifiers have been assigned then the digital audio recordings will be destroyed. All data will be retained in accordance with current Data Protection Regulations.

### 7.6 RANDOMISATION AND MASKING

Participants will complete informed consent/assent and baseline questionnaires online within LifeGuide software. Eligible participants will then be randomised to the intervention group or control group by the software.

Blinding of participants is not possible. Participants will be informed online as to which group they have been allocated to immediately. The trial statisticians will remain masked to treatment allocation until all analysis code has been written and checked. It will not be possible for all members of the study team to remain masked to treatment allocation, particularly those carrying out process evaluation interviews and those assisting with identifying participants for these interviews. Randomisation will be carried out in blocks of 4 and 6 and stratified by gender, age (13 to 15 years vs. 16 to 25 years) and recruitment source (general practice or social media/pharmacy/schools/colleges/community recruitment).

### 8 SAFETY

No specific therapeutic intervention is proposed. This programme consists of online information to support acne self-management. We will not collect adverse events. If serious adverse events are reported by study participants and are judged to be related to using the intervention these will be reported to the Sponsor and HRA.

### 8.1 SAFEGUARDING

Steps have been taken to ensure safeguarding for both the participants and the researchers. A lone worker policy is in place for the interviewers to follow prior to fieldwork. A safeguarding policy is in place and researchers will have a copy in their interview pack to follow in case of concern arising during interviews with participants.

The participant information sheet advises potential participants that confidentiality may have to be breached if there is a disclosure relating to serious harm, abuse and/or other safeguarding concerns.

### 9 FEASIBILITY TRIAL

During the feasibility trial all aspects of the protocol will be assessed. This will include (but may not be limited to): recruitment, retention and completion of questionnaires, randomisation, and participant engagement in accessing the intervention. The feasibility trial will allow for any necessary changes to the intervention to be addressed, and to build programme theory to test in the process evaluation of the full-scale trial.

If no changes are made to the intervention or to any trial procedures then data from the participants in the feasibility trial may be included in numbers for the full-scale trial, to avoid recruiting more participants than necessary overall, using established methods for decision-making on whether feasibility or pilot data can be carried forward without compromising trial integrity. [42]

### **10 INTERNAL PILOT** AND FULL-SCALE TRIAL

If the feasibility trial shows that no changes are needed to the intervention or to trial procedures then we will progress straight to full scale trial. However, if substantial changes are needed to the intervention or trial procedures (for instance recruitment strategy) then relevant changes will be made and submitted for amendment to ethics approval if necessary. Once the full scale trial commences, no further changes will be made to the intervention.

### **11 STUDY PROCEDURES**

### 11.1 STUDY PROCEDURES AND FOLLOW UP

Study procedures for both feasibility and full-scale trials will be carried out online through the online interventions (LifeGuide software). Participants wishing to take part in the study will provide consent and complete an online baseline questionnaire as indicated in the schedule of observations before being randomised to either the intervention group or the control group. Participants in the intervention group will then have access to the intervention. Participants in the control group will be signposted to the NHS advice page on acne and given access to the intervention website after 52 week follow-up. All participants will then have access to the maintained by the study team until at least the end of the grant funding.

All participants will be asked to complete questionnaires as described above. Automated emails and text messages will be sent to notify participants when their follow-up questionnaires are available for completion.

Reminder emails and messages will be sent to non-responders (no more than 3 messages), followed by reminder telephone calls to invite participants to complete follow-up questions over the phone (no more than 3 calls).

### **11.2 DEVIATIONS AND SERIOUS BREACHES**

Any study protocol deviations, non-compliance, or breaches are departure from the approved protocol. These will be adequately documented in the deviation log and reported to the Chief Investigator and Sponsor immediately. The Sponsor will then advise of and/or undertake any corrective and preventative actions as required. Deviations from the protocol which are found to frequently recur are not acceptable, will require immediate action and could potentially be classified as a serious breach.

All serious protocol deviations, non-compliance or serious breaches of Good Clinical Practice and /or the study protocol will immediately be reported to the relevant Research Ethics Committee (REC).

### 11.3 WITHDRAWAL

The participant or legal representative is free to withdraw consent from the study at any time without providing a reason. If a participant initially consents but subsequently withdraws from the trial, a clear distinction will be made where possible as to what aspect of the trial the participant is withdrawing from. These aspects could be:

- 1. Withdrawal from trial intervention
- 2. Withdrawal from further study follow-up
- 3. Withdrawal from entire study and does not want data to be used.

The study team will explain the value of remaining in study follow-up and allowing this data to be used for trial purposes. Where possible, participants who have withdrawn from the study will remain in follow-up as per the trial schedule. If participants additionally withdraw consent for further follow up, they will not be followed up further.

Details of study discontinuation (date, reason if known) will be recorded in the trial management database.

### **12 INTERVENTION AND GROUP DETAILS**

Participants will be randomised to either

- 1) online intervention plus usual care or
- 2) signposting to NHS information plus usual care, plus access to the intervention after 52 week follow-up

Participants in both groups will continue to receive their usual medical advice and prescriptions provided by their GP surgery and other healthcare services. They will not be prevented from seeking additional online support.

### **13 STATISTICS AND DATA ANALYSES**

### 13.1 METHOD OF RANDOMISATION

Participants will complete informed consent, baseline questionnaires online and then will be randomised within the software. It will not be possible to blind participants to their allocation group, and it will not be possible to blind research team carrying out process evaluation interviews, but the trial statisticians will remain blinded. Randomisation will be carried out in random permuted blocks of 4 and 6 and stratified by gender, age (13 to 15 years vs. 16 to 25 years) and recruitment source (general practice or social media /pharmacy /schools /colleges /community recruitment).

### 13.2 SAMPLE SIZE

Based on comparison of the Acne-QoL symptoms subscale at 12 weeks, power 90%, alpha 0.05 and seeking a standardised effect size of 0.30 (standard deviation 5.8), 470 participants are needed. Allowing for 20% loss to follow up gives a total 588 participants (294 per arm). An effect size of 0.30 equates to a difference in Acne-QoL symptoms subscale between arms of 1.74 points, meaning we would be conservatively powered to detect the published minimal clinically reported difference for Acne-QoL, which is a difference of 2 points on the symptom subscale.[28, 29] A standard deviation 5.8 aligns with trials with a similar patient group.[29] Based on recruitment to SAFA (HTA16/13/02) this sample size is achievable in the proposed timescale.

For the full-scale trial, we anticipate that we will need to recruit up to 100 practices to randomise 300 participants (half of total as the remainder to be recruited through social media and other sources), although this will be reviewed following the feasibility trial.

### 13.3 STATISTICAL ANALYSIS PLAN (SAP)

A detailed statistical analysis plan will be developed prior to the final analysis of the trial, however, the main features of the plan are discussed below. Data will be reported and presented according to the revised CONSORT (Consolidated Standards of Reporting Trials) Statement. All results will be reported with 95% confidence intervals and all models will control for the baseline covariates set out in the primary analysis. Any subgroup analyses will be planned and pre-specified in the statistical analysis plan.

### Feasibility trial

The analysis of the feasibility study data will be descriptive, focusing on the recruitment and retention rates, as well as participant characteristics to ensure we are recruiting a diverse sample. We will explore the levels of missing data both at the questionnaire and individual questionnaire item levels to explore the acceptability of the outcome measures. We will estimate the standard deviation for our primary outcome and revisit our sample size assumptions for the full-scale trial with oversight committees. If no substantial changes to intervention, recruitment or questionnaires are needed following the feasibility trial then data

from feasibility trial participants will contribute to internal pilot trial and full-scale trial analyses.

#### **Internal pilot**

The analysis of the internal pilot will be descriptive and again discussed with oversight committees. If no substantial changes to recruitment are needed following the internal pilot then data from these participants will contribute to the full-scale trial analyses. We propose the following progression criteria from internal pilot to full-scale trial:

- Green: if recruitment exceeds 70% of the recruitment predicted the full-scale trial will continue unchanged.
- Amber: if recruitment is 50-70% of the recruitment predicted this will be discussed with oversight committees to consider altering recruitment strategy then proceed with monthly recruitment updates.
- Red: if recruitment is less than 50% predicted this will be discussed with oversight committees and funder. Unless there is a credible plan to increase recruitment, progression to the full-scale trial should be halted.

#### Full-scale trial

For our primary outcome, we will use a linear regression model to analyse Acne-QoL symptom subscale at 12 weeks. All analyses will control for key baseline covariates and stratification factors. The full model will be set out in the statistical analysis plan prior to database lock. A 95% confidence interval for the mean difference between arms in Acne-QoL symptom subscale at 12 weeks will be calculated and reported. The frequency and pattern of missing data will be examined and a multiple imputation model will be used as a sensitivity analysis if appropriate.

As a secondary analysis we will also assess Acne-QoL symptom subscale and other Acne-QoL subscales with a repeated measures approach, using a multilevel mixed model allowing for observations (level 1) at all timepoints (12 weeks, 24 weeks, 36 weeks and 52 weeks) nested within participants (level 2) and adjusting for the same covariates as the primary analysis.

We will also use the same analysis method as the primary outcome to summarise Acne-QoL symptom subscale at each of the other time points and for the other Acne-QoL subscales (self-perception, role-emotional and role-social) and total score.

Secondary outcomes will be modelled using a regression modelling approach with a distribution appropriate to the outcome: linear regression for continuous outcomes or quantile regression if the assumptions for linear modelling are not met, logistic regression for binary outcomes, and a suitable count distribution such as Poisson or negative binomial for count data. All results will be reported with 95% confidence intervals and all models will control for the baseline covariates set out in the primary analysis. No pre-specified subgroup analyses are planned.

### **14 NESTED HEALTH ECONOMIC EVALUATION**

The economic evidence base for acne is under-researched, particularly in the UK context.[43] This programme grant will provide an opportunity to test and apply resource use data collection tools and preference-based measures with people with acne.

### Feasibility trial

Within the feasibility trial phase the focus will be on:

- Testing the use of an (online) aide memoire diary to help capture resource use and self-report questionnaire for collection of resource use data. Completion rates will be assessed and potentially important resource items identified.
- Seeking participants' thoughts on the preference-based measures (EQ-5D-5L[44] and Short Warwick Edinburgh Mental Well-being Scale (SWEMWBS)[38]) and their relevance in acne to help inform how to measure outcomes for use in the economic evaluation to be conducted alongside the full-scale trial.

It is important to explore the most appropriate way to measure outcome for use in the economic evaluation. Despite limited published evidence suggesting support for the use of the EQ-5D in acne, [44, 45] a recent acne trial [39] found a large proportion of the participants started the trial in perfect health on the EQ-5D (46% in the intervention group and 43% in the control group) and thus the EQ-5D had no capacity to measure any benefit that might arise from the intervention. Four of the five dimensions on the EQ-5D measure physical health and so the applicability of the EQ-5D and SWEMWBS will depend, in part, on the importance of mental well-being as opposed to physical health when valuing outcomes for acne.

### Full-scale trial

The economic aims within the full-scale trial are:

- Describe the types of resource used for acne within the study sample.
- Estimate resource use and costs in people with acne in the intervention group compared to the usual care group.
- Estimate the Quality-Adjusted Life Years (QALYs)/Mental Wellbeing-Adjusted Life Years (mWALYs) in people living with acne in both arms.
- Undertake a cost-utility analysis to inform decision-making about whether the online intervention should be promoted for acne care in the NHS.

A within trial economic evaluation will estimate whether the online intervention is costeffective compared to usual care from an NHS and personal social services perspective, and separately from a participant/family perspective as appropriate. We will estimate cost of the intervention and wider health care costs. Data on intervention resource use will be collected by the trial team whilst wider health resource use, in particular acne-related prescriptions, service use and out of pocket costs and productivity costs incurred by participants (or their family) will be collected through self-complete online questionnaires at baseline, 12, 24, 36 and 52 weeks with participants also given (online) resource use aide memoire diaries to use between data collection points to aid recall. Data collection will be kept minimal, and the online format will allow for skipping of sections where no resources have been used.

Depending on the outcome of the feasibility work with respect to outcome measurement we will measure either EQ-5D-5L and/or Short Warwick Edinburgh Mental Well-being Scale (SWEMWBS) at baseline, 12, 24, 36 and 52 weeks in the full-scale trial. Health-related quality of life if measured using EQ-5D will be done so in line with NICE recommendations.[37] EQ-5D-5L scores will be converted to utility scores in line with current recommendations and will be used to estimate Quality-Adjusted Life Years (QALYs) for the trial period.[46] The Short Warwick Edinburgh Mental Well-being Scale (SWEMWBS) consists of seven questions and for which a UK preference-based value set has been published enabling the estimation of Mental Well-being Adjusted Life Years (MWALY).[38]

If the EQ-5D is used in the full-scale trial, this data will be used to explore the measurement properties (validity, responsiveness, practicality) of the EQ-5D for mild and moderate acne in order to inform future studies and the core outcome set for acne.[30] A cost utility analysis will be performed using accepted methods to inform judgements about the value for money afforded by the online intervention. Using information on costs and benefits, regression analysis will estimate the incremental cost, incremental benefit and incremental cost effectiveness of the online intervention compared to usual care alone. If one arm is clearly dominant (less costly and more effective) a recommendation can be made on this basis. If non-dominance occurs (costs are greater and the intervention more effective or if the intervention is cheaper and less effective), an incremental cost-effectiveness ratio (ICER) will be estimated so that a judgement about value for money can be made[47, 48] Sensitivity analyses will be undertaken to explore the impact of key uncertainties in the analysis. These will be specified in advance in the Health Economics Analysis Plan (HEAP) which will be written and reviewed before the trial data base is locked.

The value of undertaking long-term modelling will be assessed with a view to seeking further funding opportunities to undertake such work should such an approach seem useful. This is likely should the intervention prove effective in the trial but not cost-effective as the long-term cost-effectiveness may be different if establishing effective early acne treatment avoids sequelae, such as scarring, and reduces referrals to secondary care.

### **15 NESTED PROCESS EVALUATION STUDY**

A mixed-methods process evaluation study will explore how participants engage with, and experience, the Acne Care Online trial and intervention and will provide an understanding of the mechanisms through which outcomes are achieved. Participants from both intervention and control arms of the trial will be included in all process analyses to provide comparative insight into the experiences and outcomes of those managing their acne without the Acne Care Online intervention.

### **Feasibility trial**

Within the feasibility trial phase the focus will be on using the insights from these process analyses to:

- Identify any further changes required to intervention content and/or trial processes
- Ensure data collection measures are feasible, accessible and sufficient to provide the data required
- Refine the intervention logic model to inform hypotheses to test about anticipated mechanisms of action within the full-scale trial

#### Full-scale trial

Within the fully-powered trial, the focus will be on using the insights from these process analyses to:

- Understand how participants engage with, and experience, the trial and intervention
- Understand factors that mediate and moderate outcomes for participants
- Provide further contextual understanding of the trial outcomes

### 15.1 QUANTITATIVE PROCESS STUDY

#### 15.1.1 Data collection

Quantitative process data will be collected from all participants in each phase of the study via LifeGuide software. This data will be collected via self-report questionnaires and by measures of intervention usage automatically recorded in the software (with informed participant consent).

Self-reported measures of intervention use that participants are asked to complete at baseline, 12 weeks, 24, weeks, 36 weeks and 52 weeks as part of the trial data collection will allow exploration of key constructs in the quantitative process evaluation: beliefs about acne and acne treatments, perceptions of self-efficacy, outcome expectations, psychosocial impact of acne, adherence to treatments and perceived barriers to this and treatment/advice-seeking behaviours. The specific measures and their collection time-points are shown in the 'Schedule of Observations and Procedures'.

Objective measures of intervention use automatically recorded in LifeGuide software, with participant consent, will allow evaluations of intervention usage patterns, such as time spent on intervention, number of visits to the intervention website and pages visited.

### 15.1.2 Data analysis

Descriptive and inferential statistics will be used to describe the quantitative process data and to examine relationships between key variables of interest. During the feasibility trial we will refine our logic model and anticipated mechanisms of change to identify the specific relationships between variables that we plan to analyse in the full-scale trial, and the appropriate statistical methods to apply.

We will use baseline data to examine potential predictors and moderator effects of participant characteristics (eg., age, acne severity, baseline attitudes) on intervention engagement (objectively recorded detailed website usage and self-reported treatment adherence) and outcome. We will also assess and analyse hypothesised mediators of treatment adherence and

intervention outcomes; specifically changes in beliefs about treatment (PETS) as well as intervention usage.

### 15.2 QUALITATIVE PROCESS STUDY

### 15.2.1 Data collection

Qualitative process data will be collected from a sub-sample of participants from each phase of the trials. In the feasibility trial, we aim to interview up to 10 participants from each trial arm. In the full-scale trial, we aim to interview approximately 15-20 people in the intervention group and approximately 5 people in the control group (20-25 in total). We aim to conduct interviews with all participants within three months of them being randomised into the trial. We will try to achieve a maximum variation sample of included participants with regards to characteristics such as age, gender, ethnicity, acne duration and perceived severity, level of usage of the online intervention (for intervention group).

Participants will be identified and invited via their participation in the trial. As part of the trial consent process all participants signing up will be asked if they are willing to be recontacted about later participation in a qualitative interview. When the interviews are due, we will select our maximum variation sample from those participants indicating their willingness to be recontacted. These individuals will then be contacted by phone, text message or email for the study team to provide further information about the interview study and the opportunity to ask questions. They will then be asked to give informed consent online prior to the interview, or if they do not manage to complete online consent but attend for an interview then verbal consent will be recorded at the start of the interview. Where face-to-face interviews are carried out written informed consent will be sought in person. Participants will be free to withdraw at any time by contacting the research team.

For process evaluation interviews with participants aged 13-15, written consent from parent/carer will be obtained prior to seeking assent from the young person.

### 15.2.2 Data analysis

Data from the qualitative process study will be analysed, firstly, using reflexive thematic analysis to construct themes grounded in the data. This will ensure that these represent participant perspectives and have the potential to challenge or extend our understanding of the factors and mechanisms involved in intervention outcomes. Following this, we will also examine the ways in which these inductively derived themes may map onto, elaborate or diverge from our theoretical frameworks, so as to relate our context-specific insights to generalisable theoretical constructs (where possible and appropriate). The findings will be discussed and interpretations agreed between the co-investigators (including PPI representatives).

### 15.3 PROCESS EVALUATION ANALYSIS

We will triangulate findings from the quantitative and qualitative process analyses to explore and test the causal mechanisms proposed, to help inform interpretation of trial results, and determine how the interventions could be improved and how implementation into clinical practice could be facilitated.

### **16 REGULATORY**

### 16.1 CLINICAL TRIAL AUTHORISATION

This trial is not considered to be a clinical trial of a medicinal product or a medical device, so clinical trial authorisation from the UK Competent Authority the Medicines and Healthcare products Regulatory Agency (MHRA) is not applicable.

### **17 ETHICAL CONSIDERATIONS**

### 17.1 ETHICAL APPROVAL

The trial protocol has received the favourable opinion of a Research Ethics Committee - (North West - Greater Manchester East Research Ethics Committee.

An annual progress report will be submitted to the REC within 30 days of the anniversary date on which the favourable opinion was given, and annually until the trial is declared ended.

Within one year after the end of trial, the Chief Investigator will submit a final report with the results, including any publication/abstracts, to the REC.

### **18 SPONSOR**

The University of Southampton is acting as Sponsor for this trial.

### 18.1 INDEMNITY

The University of Southampton's public and professional indemnity insurance policy provides an indemnity to UoS employees for their potential liability for harm to participants during the conduct of the research. This does not in any way affect an NHS' Trust's responsibility for any clinical negligence on the part of its staff.

### 18.2 FUNDING

This study is funded by the National Institute for Health and Care Research Programme Grants for Applied Research NIHR202852. The views expressed are those of the author(s) and not necessarily those of the NIHR or the Department of Health and Social Care.

#### **GP** practices

GP practices will be reimbursed through service support payments from the NIHR CRN and research costs from the Acne Care Online study.

#### Participants

For the trial, participants will be entered into a prize draw to win one of four £50 vouchers at 12 weeks, one of four £50 vouchers at 24 weeks and one of four £50 vouchers at 52 weeks. This will be advertised to participants after they have completed baseline processes as it is an

incentive for completing the trial, rather than for signing up to the trial. Participants taking part in the nested qualitative study interviews will receive a £15 gift voucher for their time.

### **18.3 AUDITS AND INSPECTIONS**

The study may be participant to inspection and audit by the University of Southampton (under their remit as Sponsor) and other regulatory bodies to ensure adherence to the principles of GCP, Research Governance Framework for Health and Social Care, applicable contracts/agreements and national regulations.

### **19 STUDY OVERSIGHT GROUPS**

The day-to-day management of the study will be co-ordinated through the trial management team in Southampton, and oversight will be maintained by the Programme Management Group and the Programme Steering Committee.

### **19.1 PROGRAMME MANAGEMENT GROUP (PMG)**

The PMG is responsible for overseeing progress of the study, including both the clinical and practical aspects. The Acne Care Online Programme Management Group (PMG) will be the PMG for the RCT. MS and IM are the Chief Investigators for this trial and will chair the PMG. The PMG includes three public contributors with experience of acne.

The Acne Care Online PMG charter defines the membership, terms of reference, roles, responsibilities, authority, decision-making and relationships of the PMG, including the timing of meetings, frequency and format of meetings and relationships with other trial committees.

### **19.2 PROGRAMME STEERING COMMITTEE (PSC)**

The PSC act as the oversight body on behalf of the Sponsor and Funder. The Acne Care Online Programme Steering Committee (PSC) will be the TSC for the RCT. The PSC/TSC will meet in person at least yearly. The Acne Care Online PSC/TCS consists of four independent members, including the Chair. One of the four independent members is a public contributor with experience of acne.

The Acne Care Online PSC charter defines the membership, terms of reference, roles, responsibilities, authority, decision-making and relationships of the PSC, including the timing of meetings, frequency and format of meetings.

### **19.3 DATA MONITORING AND ETHICS**

The aim of the DMEC is to safeguard the interests of study participants, monitor the main outcome measures including safety and efficacy, and monitor the overall conduct of the study. These duties will be carried out by the Acne Care Online Programme Steering Committee.

### **20 RECORD RETENTION AND ARCHIVING**

Study documents will be retained in a secure location during and after the trial has finished.

The PI or delegate must maintain adequate and accurate records to enable the conduct of the study to be fully documented and the study data to be subsequently verified. After study closure the PI will maintain all source documents and study related documents. All source documents will be retained for a period of 10 years following the end of the study.

### **21 PUBLICATION POLICY**

Data for the RCT from all centres will be analysed together and published as soon as possible. All publications and presentations relating to the trial will be authorised by the Programme Management Group. Participants and participating GP surgeries will be sent a summary of results once available. The summary will also be made available on the study website in an accessible format.

### **22 REFERENCES**

- 1. Hay, R.J., et al., *The global burden of skin disease in 2010: an analysis of the prevalence and impact of skin conditions.* Journal of Investigative Dermatology, 2014. **134**(6): p. 1527-1534.
- 2. Williams, H.C., R.P. Dellavalle, and S. Garner, *Acne vulgaris*. The Lancet, 2012. **379**(9813): p. 361-372.
- 3. Hassan, J., et al., *The individual health burden of acne: appearance-related distress in male and female adolescents and adults with back, chest and facial acne.* Journal of health psychology, 2009. **14**(8): p. 1105-1118.
- 4. Bhate, K. and H. Williams, *Epidemiology of acne vulgaris*. British Journal of Dermatology, 2013. **168**(3): p. 474-485.
- 5. Ghodsi, Z., H. Orawa, and C.C. Zouboulis, *Prevalence, Severity, and Severity Risk Factors of Acne in High School Pupils: A Community-Based Study.* Journal of Investigative Dermatology, 2009. **129**(9): p. 2136-2141.
- 6. Lown, M., et al., *Prescribing of Long-term Antibiotic to Adolescents in Primary Care: a Retrospective Cohort Study*, in *Society of Academic Primary Care Annual Scientific Meeting*. 2020.
- 7. Purdy, S., J. Langston, and L. Tait, *Presentation and management of acne in primary care: a retrospective cohort study.* British Journal of General Practice, 2003. **53**: p. 5.
- 8. Shapiro, D.J., et al., *Antibiotic prescribing for adults in ambulatory care in the USA, 2007–09.* Journal of Antimicrobial Chemotherapy, 2014. **69**(1): p. 234-240.
- 9. Walsh, T.R., J. Efthimiou, and B. Dréno, Systematic review of antibiotic resistance in acne: an increasing topical and oral threat. The Lancet Infectious Diseases, 2016. **16**(3): p. e23-e33.
- 10. Layton, A., et al., A review of the European Directive for prescribing systemic isotretinoin for acne vulgaris. Journal of the European Academy of Dermatology and Venereology, 2006. **20**(7): p. 773-776.
- 11. Xu, J., et al., *Management of acne vulgaris: summary of NICE guidance.* bmj, 2021. **374**.
- 12. Dréno, B., et al., *Large-scale worldwide observational study of adherence with acne therapy.* International journal of dermatology, 2010. **49**(4): p. 448-456.
- 13. Thiboutot, D., B. Dréno, and A. Layton, *Acne counseling to improve adherence*. Cutis, 2008. **81**(1): p. 81-86.
- 14. McNiven, A., *Disease, illness, affliction? Don't know': Ambivalence and ambiguity in the narratives of young people about having acne.* Health, 2019. **23**(3): p. 273-288.

- 15. Desai, K.P., et al., *Help-seeking behaviours, opportunistic treatment and psychological implications of adolescent acne: cross-sectional studies in schools and hospital outpatient departments in the UK.* BMJ Open, 2017. **7**(9): p. e016964.
- Ip, A., et al., Young people's perceptions of acne and acne treatments: secondary analysis of qualitative interview data. British Journal of Dermatology, 2020.
   183(2): p. 349-356.
- 17. Francis, N.A., et al., *The Management of Acne Vulgaris in Primary Care: A cohort study of consulting and prescribing patterns using CPRD.* British Journal of Dermatology, 2016: p. n/a-n/a.
- Platt, D., et al., GPs' perspectives on acne management in primary care: a qualitative interview study. British Journal of General Practice, 2021. 71(702): p. e78-e84.
- 19. NHSEngland. *Guidance on conditions for which over the counter items should not routinely be prescribed in primary care*. 2018 [cited 2020 16 November]; Available from: <u>https://www.england.nhs.uk/medicines-2/conditions-for-which-over-the-counter-items-should-not-routinely-be-prescribed/</u>.
- 20. Layton, A., C. Henderson, and W. Cunliffe, *A clinical evaluation of acne scarring and its incidence.* Clinical and experimental dermatology, 1994. **19**(4): p. 303-308.
- 21. Tuchayi, S.M., et al., *Interventions to increase adherence to acne treatment*. Patient preference and adherence, 2016. **10**: p. 2091.
- 22. Koch, P.E., et al., *Educating adolescents about acne vulgaris: a comparison of written handouts with audiovisual computerized presentations.* Archives of dermatology, 2008. **144**(2): p. 208-214.
- 23. Tucker, R., et al., *The development and feasibility testing of a community pharmacy-based guidelines for the assessment and management of acne vulgaris.* 2019.
- 24. Webb, T., et al., Using the internet to promote health behavior change: a systematic review and meta-analysis of the impact of theoretical basis, use of behavior change techniques, and mode of delivery on efficacy. Journal of medical Internet research, 2010. **12**(1): p. e4.
- 25. Murray, E., et al., *Interactive Health Communication Applications for people with chronic disease.* Cochrane database of systematic reviews, 2004(4).
- 26. Hornsey, S., et al., *Patient-reported outcome measures for acne: a mixed-methods validation study (acne PROMs).* BMJ Open, 2021. **11**(3): p. e034047.
- 27. Martin, A.R., et al., *Health-related quality of life among patients with facial acne: assessment of a new acne-specific questionnaire.* Clinical and Experimental Dermatology, 2001. **26**: p. 380-385.
- McLeod, L., D., et al., Evaluating Minimal Clinically Important Differences for the Acne-Specific Quality of Life Questionnaire. Pharmacoeconomics, 2003. 21(15): p. 1069-1079.
- 29. Fehnel, S.E., et al., *Responsiveness of the Acne-Specific Quality of Life Questionnaire (Acne-QoL) to treatment for acne vulgaris in placebo-controlled clinical trials.* Quality of Life Research, 2002. **11**(8): p. 809-816.
- 30. ACORN. *Welcome to ACORN*. [cited 2017; Available from: <u>https://sites.psu.edu/acnecoreoutcomes/</u>.
- 31. Thiboutot, D., et al., *Assessing effectiveness in acne clinical trials: steps towards a core outcome measure set.* British Journal of Dermatology, 2019. **181**(4): p. 700-706.
- 32. Ozolins, M., et al., *Randomised controlled multiple treatment comparison to provide a cost-effectiveness rationale for the selection of antimicrobial therapy in acne*, in *Health Technology Assessment HTA* 2005, NHS R&D HTA Programme.

- 33. Kirby, S., M. Donovan-Hall, and L. Yardley, *Measuring barriers to adherence:* validation of the problematic experiences of therapy scale. Disability and rehabilitation, 2014. **36**(22): p. 1924-1929.
- 34. Howie, J., et al., A comparison of a patient enablement instrument (PEI) against two established satisfaction scales as an outcome measure of primary care consultations. Family practice, 1998. **15**(2): p. 165-171.
- 35. Broadbent, E., et al., *The Brief Illness Perception Questionnaire.* Journal of Psychosomatic Research, 2006. **60**(6): p. 631-637.
- 36. Löwe, B., et al., A 4-item measure of depression and anxiety: validation and standardization of the Patient Health Questionnaire-4 (PHQ-4) in the general population. Journal of affective disorders, 2010. **122**(1-2): p. 86-95.
- 37. NICE, Position statement on use of the EQ-5D-5L valuation set. 2017.
- Yiu, H.H.E., et al., Derivation of a UK preference-based value set for the Short Warwick-Edinburgh Mental Well-being Scale (SWEMWBS) to allow estimation of Mental Well-being Adjusted Life Years (MWALYs). Social Science & Medicine, 2023. 327: p. 115928.
- 39. Santer, M., et al., *Effectiveness of spironolactone for women with acne vulgaris* (SAFA) in England and Wales: pragmatic, multicentre, phase 3, double blind, randomised controlled trial. BMJ, 2023. **381**: p. e074349.
- 40. Franklin, M. and J. Thorn, Self-reported and routinely collected electronic healthcare resource-use data for trial-based economic evaluations: the current state of play in England and considerations for the future. BMC medical research methodology, 2019. **19**(1): p. 1-13.
- 41. Noben, C.Y., et al., *The exchangeability of self-reports and administrative health care resource use measurements: assessement of the methodological reporting quality.* Journal of clinical epidemiology, 2016. **74**: p. 93-106. e2.
- 42. Charlesworth, G., et al., *Acceptance checklist for clinical effectiveness pilot trials: a systematic approach.* BMC Medical Research Methodology, 2013. **13**(1): p. 78.
- 43. Layton, A., D. Thiboutot, and J. Tan, *Reviewing the Global Burden of acne: how could we improve care to reduce the burden?* British Journal of Dermatology, 2020.
- 44. Yang, Y., J. Brazier, and L. Longworth, *EQ-5D in skin conditions: an assessment of validity and responsiveness.* The European Journal of Health Economics, 2015. **16**(9): p. 927-939.
- 45. Klassen, A.F., J.N. Newton, and E. Mallon, *Measuring quality of life in people referred for specialist care of acne: Comparing generic and disease-specific measures.* Journal of the American Academy of Dermatology, 2000. **43**(2, Part 1): p. 229-233.
- 46. Manca, A., N. Hawkins, and M.J. Sculpher, *Estimating mean QALYs in trial*based cost-effectiveness analysis: the importance of controlling for baseline utility. Health Econ, 2005. **14**(5): p. 487-96.
- 47. Ramsey, S.D., et al., *Cost-effectiveness analysis alongside clinical trials II-An ISPOR Good Research Practices Task Force report.* Value Health, 2015. **18**(2): p. 161-72.
- 48. Drummond, M., F., et al., *Methods for the economic evaluation of health care programmes*. 4th edition ed. 2015, Oxford: Oxford University Press.